\documentclass[12pt,letterpaper]{article}

\usepackage{graphicx,color}
\usepackage{natbib}
\usepackage{amsmath,amsfonts,amsthm,amssymb,euscript,mathrsfs,url,bbm}
\usepackage{algorithmic, stmaryrd}
\usepackage[tight]{subfigure}
\usepackage{psfrag}
\usepackage{multirow}
\usepackage{fancyhdr,lastpage}

% some formatting from http://www.tedpavlic.com/post_homework_tex_example.php:
% In case you need to adjust margins:
\topmargin=-0.45in      %
\evensidemargin=-0.1in     %
\oddsidemargin=-0.4in      %
\textwidth=7.3in        %
\textheight=9.0in       %
\headsep=0.25in         %
\headheight=30pt

% Setup the header and footer
\pagestyle{fancy}
\lhead{CS 390: Topics in Artificial Intelligence\\\hmwkTitle:\ Due: October 5, 2016}
\cfoot{}
\rfoot{Page\ \thepage\ of\ \protect\pageref{LastPage}}
\renewcommand\headrulewidth{0.4pt}
\renewcommand\footrulewidth{0.4pt}

\newcommand{\R}{\mathbb{R}}
\newcommand{\Z}{\mathbb{Z}}
\newcommand{\N}{\mathbb{N}}
\newcommand{\Q}{\mathbb{Q}}

\newcommand{\var}{\mathop{\mbox{Var}}}
\newcommand{\cov}{\mathop{\mbox{Cov}}}
\newcommand{\grad}{\nabla}

\newcommand{\sCol}[1]{\mathscr{#1}}
\newcommand{\sCmp}[1]{#1^{\text{c}}}
\newcommand{\sigF}{\sigma\text{-field}}

\newcommand{\iidsim}{\stackrel{\mathrm{iid}}{\sim}}

\DeclareMathOperator{\leb}{Leb}
\DeclareMathOperator{\cl}{cl}
\DeclareMathOperator{\Ber}{Ber}
\DeclareMathOperator{\Poi}{Poi}
\DeclareMathOperator{\Gam}{Gam}
\DeclareMathOperator{\Bin}{Bin}
\DeclareMathOperator{\IG}{IG}
\DeclareMathOperator{\Max}{Max}
\DeclareMathOperator{\Exp}{Exp}
\DeclareMathOperator{\Nor}{Normal}
\DeclareMathOperator{\Uni}{Unif}
\DeclareMathOperator{\Par}{Pareto}
\DeclareMathOperator{\Dir}{Dir}
\DeclareMathOperator{\Bet}{Beta}
\DeclareMathOperator{\Cau}{Cau}
\DeclareMathOperator{\TN}{TN}
\DeclareMathOperator{\Null}{Null}

\DeclareMathOperator{\E}{E}

\newcommand{\probnumber}[1]{\noindent\textbf{(#1)}}

% Homework Specific Information
\newcommand{\hmwkTitle}{Homework 3}
\begin{document}

%%%%%%%%%%%%%%%%%
%%%%%% PAPER 1 %%%%%%
%%%%%%%%%%%%%%%%%

\noindent {\color{blue}{\textbf{\underline{Paper 1, ``Data clustering: 50 years beyond K-means''}}}} \\
	
\noindent {\color{blue}{Briefly summarize (2-3 sentences each):}}
	
\begin{enumerate}
	
	{\color{blue}{\item Motivation/purpose of the paper (why they did it)}}
	
	% remove these vspaces (vertical space) and add your own text
	\vspace{5mm}
	
	{\color{blue}{\item Methodology (how they did it)}}
	\vspace{5mm}
	
	{\color{blue}{\item Results (what they found)}}
	\vspace{5mm}
	
	{\color{blue}{\item Conclusions (what it means)}}
	\vspace{5mm}
	
\end{enumerate}
	
\noindent {\color{blue}{What parts of the paper did you spend the most time on?  What parts seemed skimmable?  Did you read the sections in order?}}
\vspace{15mm}
	
\noindent {\color{blue}{What vocabulary from the paper were you unfamiliar with?}}
\vspace{15mm}
	
\noindent {\color{blue}{What strategies did you use to learn this new vocabulary?}}
	
\vspace{15mm}
	
\noindent {\color{blue}{How would you rate the paper in terms of:}}
	
\begin{itemize}
	
	{\color{blue}{\item Well-motivated scientific question}}
	\vspace{5mm}
	
	{\color{blue}{\item Clear writing}}
	\vspace{5mm}
	
	{\color{blue}{\item Novel and/or well-chosen method}}
	\vspace{5mm}
	
	{\color{blue}{\item Scientific conclusions}}
	\vspace{5mm}
	
\end{itemize}
	
\noindent {\color{blue}{\underline{Equation:}}} (example below, replace)

% note double dollar sign ($$) for centered math mode on it's own line, vs. single dollar sign ($) for in-line math mode
$$\int_0^\infty e^{-x^2} dx=\frac{\sqrt{\pi}}{2}$$

\vspace{5mm}
	

\hfill {\scriptsize {\color{blue}{Reference: ``How to Read a Scientific Article,'' Purugganan and Hewitt (2004)}}}
\newpage

%%%%%%%%%%%%%%%%%
%%%%%% PAPER 2 %%%%%%
%%%%%%%%%%%%%%%%%

\noindent {\color{blue}{\textbf{\underline{Paper 2, ``Dynamic hierarchical algorithms for document clustering''}}}} \\ % note use of quotes
	
	\noindent {\color{blue}{Briefly summarize (2-3 sentences each):}}
	
\begin{enumerate}
	
	{\color{blue}{\item Motivation/purpose of the paper (why they did it)}}
	
	% remove these vspaces (vertical space) and add your own text
	\vspace{5mm}
	
	{\color{blue}{\item Methodology (how they did it)}}
	\vspace{5mm}
	
	{\color{blue}{\item Results (what they found)}}
	\vspace{5mm}
	
	{\color{blue}{\item Conclusions (what it means)}}
	\vspace{5mm}
	
\end{enumerate}
	
\noindent {\color{blue}{What parts of the paper did you spend the most time on?  What parts seemed skimmable?  Did you read the sections in order?}}
\vspace{15mm}
	
\noindent {\color{blue}{What vocabulary from the paper were you unfamiliar with?}}
\vspace{15mm}
	
\noindent {\color{blue}{What strategies did you use to learn this new vocabulary?}}
	
\vspace{15mm}
	
\noindent {\color{blue}{How would you rate the paper in terms of:}}
	
\begin{itemize}
	
	{\color{blue}{\item Well-motivated scientific question}}
	\vspace{5mm}
	
	{\color{blue}{\item Clear writing}}
	\vspace{5mm}
	
	{\color{blue}{\item Novel and/or well-chosen method}}
	\vspace{5mm}
	
	{\color{blue}{\item Scientific conclusions}}
	\vspace{5mm}
	
\end{itemize}
	
\noindent {\color{blue}{\underline{Equation:}}} (example below, replace)

$$\int_0^\infty e^{-x^2} dx=\frac{\sqrt{\pi}}{2}$$

\vspace{5mm}
	

\hfill {\scriptsize {\color{blue}{Reference: ``How to Read a Scientific Article,'' Purugganan and Hewitt (2004)}}}
\newpage

	
\normalsize
	
{\color{blue}{\underline{Figure and summary:}}} (uncomment below)
	
%\begin{figure}[h!]
%	\begin{center}
%	\includegraphics[width=120mm]{myfile.pdf}
%	\end{center}
%\end{figure}

\end{document}
